

Contents lists available at ScienceDirect

Applied Energy



journal homepage: www.elsevier.com/locate/apenergy

BESS and the ancillary services markets: A symbiosis yet? Impact of market design on performance

Giuliano Rancilio^{*}, Filippo Bovera, Matteo Spiller, Marco Merlo, Maurizio Delfanti

Dipartimento di Energia, Politecnico di Milano, Via Lambruschini 4, I-20156 Milan, Italy

HIGHLIGHTS

GRAPHICAL ABSTRACT

- A comprehensive modelling suite modelling BESS vs balancing market compatibility.
- Testing regulatory barriers: symmetry, gate closure timing, dynamic services.
- Asymmetric provision of services widely improves reliability and economics.
- Revenue stacking: more selfconsumption (+10%) in synergy with ancillary services.
- Dynamic services require short distance to delivery and allow larger delivery period.



ARTICLE INFO

Keywords: Energy storage Electricity markets Balancing Ancillary services Policy BESS

ABSTRACT

The evolution of ancillary services markets (ASM) and balancing products is ongoing. The aim of the evolution is to integrate the products over the national boundaries and to open the ASM to distributed energy resources (DERs). Among DERs, battery energy storage systems (BESS) are increasing their importance. In this work, we investigate by means of numerical simulations the effect of different evolutions in the regulatory framework on the performance of a BESS providing ancillary services. The analyzed regulatory barriers are selected based on ongoing evolution in EU market design. The following parameters are involved: power vs energy-intensive

Abbreviations: ACE, Area Control Error; aFRR, automatic Frequency Restoration Reserve; ASM, Ancillary Services Market; AUX, Auxiliaries; BESS, Battery Energy Storage System; BM, Balancing Market; BtM, Behind-the-Meter; CHP, Cogenerative Heat and Power plant; DAM, Day-Ahead Market; DER, Distributed Energy Resource; DtD, Distance to Delivery; E/P, Energy-to-Power ratio; EBGL, Electricity Balancing Guidelines; En, Nominal Energy; ESS, Energy Storage System; EU, European Union; FtM, Front-of-the-Meter; FRCE, Frequency Restoration Control Error; GCT, Gate Closure Time; HVAC, Heating Ventilation Air Conditioning; IM, Intraday Market; KPI, Key Performance Indicator; kW, kilowatt; MAE, Mean Absolute Error; mFRR, manual Frequency Restoration Reserve; MW, Megawatt; MWh, Megawatt-hour; NP, Nonprovision penalty; Pn, Nominal Power; PV, Photovoltaic; qty, quantity; REF, Reference; RES, Renewable Energy Sources; RMSE, Root Mean Squared Error; RR, Replacement Reserve; SC, Self-consumption; SCADA, Supervisory Control And Data Acquisition; SoC, State-of-Charge; TD, Time Definition; TERRE, Trans European Replacement Reserves Exchange; TIDE, Testo Integrato del Dispacciamento Elettrico; TSO, Transmission System Operator; US, United States; η_{EESS}, BESS efficiency.

* Corresponding author.

E-mail address: giuliano.rancilio@polimi.it (G. Rancilio).

https://doi.org/10.1016/j.apenergy.2024.124153

Received 24 April 2024; Received in revised form 11 July 2024; Accepted 5 August 2024 Available online 13 August 2024

0306-2619/© 2024 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

services, symmetry of procurement, time definition and distance to delivery. The considered case study is a BESS associated to a large-scale energy district including load, a cogeneration plant, and a PV plant. Results are given in terms of energy flows, economics, operational efficiency, and reliability of service provision. Where both reliability, provision of large flexibility volumes, and good economic performance are achieved, we say that there is a symbiosis between BESS and the markets. This way, the best ASM arrangements abating regulatory barriers for BESS are defined.

1. Introduction

While many power systems are committed to increase (and rapidly increasing) the renewable energy sources (RES) penetration in the generation mix [1], some studies and evidence show that a larger RES penetration could increase electricity balancing costs [2–4]. In case the rise of RES is supported by improvements in balancing mechanisms, markets and technologies, the cost enhancement will likely be mitigated [5–7].

Programmable, non-fossil, flexible technologies such as energy storage systems (ESS) are considered among the key enablers for reaching high (to 100%) RES penetration [8]. Indeed, they can grant flexibility and balancing for the system via the provision of ancillary services. Ancillary services and ancillary services markets (ASM), originally built for conventional large-scale generation, should undergo a regulatory review to allow efficient and effective participation of distributed energy resources (DERs), including ESS [9,10]. In other words, regulation should be able to abate regulatory barriers [11]. When, indeed, the ESS are given the possibility to effectively provide ancillary services, they see the ASM as the core business [12], hence they are interested in offering flexibility at a competitive price, eventually lowering the system prices [13].

The research so far investigated regulatory barriers on electricity markets in different ways. Some studies [11,14] have highlighted the nature of regulatory barriers: they are not always explicit obstacles to the deployment of a technology or prohibition to enter a market. Instead, they can rely on a missing definition, on disregarding of a peculiarity of a technology, on the impossibility of capturing different revenue streams. Regulatory barriers have been categorized [15], and systematic reviews have been performed to understand which of the best trends are to follow [10]. The best practices have been highlighted and the possible lessons have been gathered [9,16,17]. Previous studies have generally proposed qualitative assessment of the impact of an ASM redesign. Some studies have proposed model-based quantitative analysis of the impact of ASM or Balancing Market (BM) redesign on the technoeconomic performance of new resources. These studies focus on a limited set of parameters [18,19], generally including or focusing on the pricing rules [20,21]. Other studies are constrained by the existing rules and services, hence they focus more on the best control strategies for existing market layouts or on a comparison between them [22-27], rather than market design evolution. Few propose as a result a grid code reform [28]. In the found studies, no more than two parameters are considered and there is not a wide sensitivity analysis on them. In general, a comprehensive and quantitative analysis assessing the compatibility of new resources with the wide range of possibilities for balancing products and markets redesign is missing. In fact, usually studies focus on the impact that BESS can have on the grid [28-30], but less on the impact that a system reform can have on BESS performance.

BESS modelling-wise, several studies are available. In the past, models were focused on the battery and the cell, modelling the electrochemical section with high accuracy [31]. For utility-scale BESS, it gets clearer and clearer that studies on the techno-economic performance of a storage system must relate with operation and existing applications [32]. Recently, operational models are proposed encompassing the whole BESS and featuring a trade-off between accuracy and computational effort, thus allowing simulations on the mid to long period and on different applications [33,34]. Among these studies,

several show that the battery is not the main source of losses in a BESS and highlight the weight of heating and cooling on plant energy losses/ inefficiencies [35–37]. A model for analysing grid-connected, largescale BESS is proposed in [38], featuring battery performance, power conversion system and auxiliary system to return BESS efficiency in market operation.

The aim of this work is quantitatively analysing the issue of ESS vs ASM compatibility, focusing on battery ESS (BESS) and modularly investigating the possibilities for redesigning balancing products. A set of simulations are performed to assess the provision of standard balancing products (thus compatible with the EU framework [39]) in different market designs by an energy district including BESS and RES. Results are given in terms of reliability of provision, effectiveness of provision (how much flexibility is provided), energy efficiency during the provision, and compatibility with behind-the-meter (BtM) services (i.e., Multiservice perspective). These quantities are then considered as aggregate to check whether there is overall economic convenience in the provision and to compare different market arrangements. Finally, to check that the proposed framework suits the analysis of real-world markets, the proposed methodology is adopted to investigate some country-specific examples of market redesign that occurred in recent years (e.g., automatic Frequency Restoration Reserve's auction scheme in Germany and balancing market reform in Italy).

The novelty of this work is proposing a modular approach in a quantitative analysis of ESS performance on ASM. This means analysing the effect of each market parameter on techno-economic performances, considering different parameters and a sound range for each of them. This allows to extensively analyze the redesign of standard balancing products. The outcomes include the overall economic evaluation of each BESS/ASM arrangement, as well as a cross comparison of specific performances, e.g., the amount of offered flexibility, the reliability of the provision, the operational BESS efficiency in each framework. The modular approach avoids binding the analysis to an existent regulatory framework. The adoption of standard balancing product can extend the validity of the results to the EU framework. The use of a multiparameter BESS model allows to consider all the main sources of inefficiency. Therefore, the work could be useful for guiding policymaking and regulatory choices while pursuing win-win arrangements for the participation of ESS to electricity balancing. Indeed, simultaneously assessing balancing services provision by the perspectives of efficiency, effectiveness, reliability, and compatibility with other services, the outcomes of the study highlight the arrangements able to gather high-level flexibility at a competitive cost.

The reminder of the paper is structured as follows. Chapter 2 details the methodology, including the market model, BESS model and control strategy for provision of ancillary services. Chapter 3 presents the case study and the layout of the simulation campaign. Chapter 4 provides techno-economic results and possible applications to evaluate realworld frameworks. Chapter 5 summarizes conclusions and policy implications.

2. Methodology

Being this work characterized by several blocks and routines, a summary of the proposed methodology is given here, including the rationale and the relation between each part (see Fig. 1), while each block is then better detailed in the following paragraphs.



Fig. 1. The block diagram for the methodology of the study, with a detail of the adopted models and their input and output.

As described in the Introduction, the literature lacks a clear identification of the optimal market arrangements for the inclusion of BESS (especially integrated with RES) in ASM and for exploiting the flexibility that can be guaranteed by these fast and precise inverter-based resources. This identification of an optimal regulatory framework, to be reliable, should be supported by quantitative analyses and evidence. A systematic review of the market structure, of the proposed services and of the traded products is already present in the literature [10], highlighting the possible evolutions to be prioritized in a qualitative way.

The goal of this work is proposing a quantitative study to better and more specifically address the integration of ESS, specifically BESS, in the markets. To assess quantitatively the regulatory barriers and check how much they can limit the battery business, a modular analysis is proposed. It aims to depict a standard framework for real-world ASM: this is to produce results that apply to real contexts, yet generalizable at least to EU. The considered features are clearly defined in art. 24 and 25 of the Electricity Balancing Guidelines (EBGL [39]): a set of parameters that are common to many products and that can be modified by the regulation while redesigning them. A subset of these parameters is modified in a range to propose the modular sensitivity analysis subject of this study. The analyzed barriers are listed in paragraph 2.1.

Two standard balancing products are tested over a serial campaign of simulations, relaxing one by one the constraints in their provision that can represent a barrier. Standard balancing products are proposed by EU regulation (in alternative to specific products) to harmonize at a continental level the provision of frequency regulation (i.e., electricity balancing). The considered products are selected since they feature different characteristics in terms of dynamics, aleatory behavior, and energy-intensity. They are presented in paragraph 2.2. Namely, these products are automatic Frequency Restoration Reserve (aFRR) and Replacement Reserve (RR) characterized using historical data [40]. In particular, the aFRR represents a power-intensive and dynamic service, controlled by an input signal variable with Frequency Restoration Control Error (FRCE). In this study, it is aimed to represent frequency response services. Oppositely, RR is an energy-intensive, static service, aimed to represent power reserve. It requires a constant power setpoint for providers. Italian data are used as input of the simulation, but the use of standard products guarantees results that can be extended to EU. The same would not be possible in the case of testing a specific product (e.g., Fast Frequency Response) [39].

The considered plant layout is the energy district whose working principles are described in paragraph 2.5.3, where the integrated BESS works as an enabler of ASM for RES and DERs. The outcomes return the acceptable (or optimal) ranges of the analyzed parameters for the integration market/BESS. As described in paragraph 2.3, a detailed BESS model featuring multiparameter and variable efficiency is adopted to improve the reliability of the results. The BESS model is adapted from study [38], and it is used to test a month of ASM participation adopting the Multiservice Strategy proposed in [41] for residential BESS, here reworked to suit the energy district and described in paragraph 2.5. Thus, the BESS is operated to provide behind-the-meter (BtM) and front-of-the-meter (FtM) services in the energy district. FtM services are based on the balancing market model illustrated in paragraph 2.4.

The results are analyzed considering several techno economic KPIs (presented in paragraph 2.6) to check the performance of the BESS and its possible correlation with the value of tested parameters. In case correlations are found, the optimal values of each parameter for enhancing BESS performance are recognized as the range where they return effectiveness, efficiency, reliability, and compatibility with other services above a threshold.

2.1. The balancing market arrangements and the analyzed barriers

As said, standard balancing products are characterized by a set of parameters. The study varies a set of these parameters in a range of interests how BESS performance in providing the service is affected. The tested parameters are presented in Fig. 2. The parameter selection has been performed starting with the suggested redesign options highlighted by the analysis presented in [10], selecting those parameters that directly impact the control strategy and the bid volumes.

- 1. The symmetry/asymmetry of products is considered in the study. Symmetry refers to the constraint of offering the same amount of upward and downward regulation. Asymmetry relaxes the constraint, e.g.: a resource can offer 1 MW upward and a different quantity downward (even 0). The aFRR and RR are proposed in the study in both configurations. In actual implementations, the balancing products are present both ways: Frequency Containment Reserve (FCR) is generally symmetric, but for Greece and Ireland; manual Frequency Restoration Reserve (mFRR) is generally asymmetric; both configurations are diffused for aFRR and RR [42].
- 2. Electricity balancing is generally pursuing a review of the services. Faster services are proposed, or more power-intensive ones, to meet the needs of the system and the peculiarities of new resources [43]. The services review is partially considered comparing two different standard products with opposed features: aFRR is power-intensive, generally characterized by a spiky power request profile (see



Fig. 2. The framework of the standard balancing product with a highlight on the tested parameters.



Fig. 3. Regulating Signal for aFRR in the Italian peninsula for the first week of 2021 [40].

Fig. 3) often changing power flow direction (withdrawing/injecting); RR is energy-intensive, usually characterized by a constant power request for the contracted time slot.

- 3. Time definition could define the minimum delivery time for a service or the duration of a market session (i.e., the distance between two market gate closures for the same product). In this case, the latter definition is adopted. This is to assess the impact of increasing the market session on limited energy resources: in principle, more sessions (i.e., lower time definition) could indicate less energy requests for the same bid power. The variation in the time definition of the products is tested by the study as depicted in Fig. 2. Shorter or longer time definitions are considered in the simulations. This is implemented via different market session durations (t_{mkt}).
- 4. The time between market gate closure and the beginning of the delivery time is often referred to as distance to delivery. It indicates how in advance a bid must be submitted to the market. Electricity markets are usually moving to occur closer to real time, both for what concerns ASM and intraday markets (IM) [44,45]. In any case, still many services are contracted a day/a week in advance [23]. Different distances to delivery are tested. This means, different time advances (t_{adv}) for estimating and bidding the flexibility.

The set of previously listed parameters/features is considered comprehensive of the evolutions of ASMs that should be more carefully analyzed and prioritized, also based on [10].

2.2. The provided ancillary services

The BESS is tested on the provision of two different standard balancing products based on the Italian data and framework. The first is aFRR: it is activated after the FCR with the aim of restoring the nominal frequency value (i.e., 50 Hz in Europe). In Italy, taken as model for the service definition, it is activated on 1-min setpoints according to the *Segnale di Livello*: it is a proportional-integral control considering as input FRCE (former ACE), correlated to frequency deviation [46]. AFRR requires a bid quantity each hour for upward and downward (P_{bidUp}(h) and P_{bidDn}(h)). If awarded, the interval [-P_{bidDn}(h), P_{bidUp}(h)] becomes the range for the setpoints related to the provision of the service, that superimposes to the exchange schedule of the resource with the grid. The *Segnale di Livello* or Regulating Signal (*S*) ranges from 0 to 100. The power to be provided each minute is defined by the following equation.

$$\begin{cases} P_{ASM}(t) = P_{bidUp}(t)^* \frac{S-50}{50} & \text{if } S \ge 50\\ P_{ASM}(t) = P_{bidDn}(t)^* \frac{S-50}{50} & \text{if } S < 50 \end{cases}$$
(2.1)

Therefore, upward service must be provided when S > 50, elsewhere the downward provision is requested. Full $P_{bidUp}(h)$ is delivered when S = 100; no power is delivered if S = 50; the full $P_{bidDn}(h)$ is absorbed with S = 0. A statistical study shows that S variability is limited in a minuteperiod (i.e., the next value is in the neighborhood of the previous one), so that a limited ramp is requested to the resources each minute. S valid for the first week of 2021, used in the study, is presented in Fig. 3. It is worth noting that, as of 2023, the regulating band of aFRR in Italy is symmetric: each resource is awarded of a $P_{bidUp}(h) = P_{bidDn}(h)$. In any case, eq. (2.1) works for both symmetric and asymmetric provision, thus allowing generalization.

The second considered ancillary service is the RR. It requests a constant power setpoint for the contracted period (e.g., 1 h), coherent with the bid quantity. Clearly, asymmetric regulating band provision is requested (i.e., the resource can offer both upward and downward and it can be awarded in either the direction or none).

Fig. 4 represents some hours of aFRR and RR provision. In both cases, the resource bids a quantity (dashed lines) for upward and a quantity for downward provision. In case the bid is awarded (shaded areas), the resource must provide the requested power (solid line). For RR, the power is constant, only depending on the awarded bid quantity. For aFRR, the requested power varies each minute based on the bid quantity and on the Regulating Signal. In the figure, asymmetric provision is depicted. The bid power for each service is defined by the control strategy as in paragraph 0.

It is worth stressing that, even if the modelling is based on Italian ancillary services, the two presented services are archetypes of standard products available in most of EU markets. E.g., the aFRR is based on FRCE, computed similarly all over EU and in US, too [47]. Instead, RR is a reserve provision with a constant setpoint requested on an hourly (or quarter-hourly) basis, it is now traded at EU level by means of the TERRE international platform [48].

2.3. The BESS modelling

In this multifaceted system, the technological modelling of the BESS is addressed by adopting an updated version of [38]. It is a multiparameter empirical model suitable for large-scale grid-connected BESS, assessing the performance of the battery, the power conversion system and the auxiliary systems. It has been validated by comparing the outcomes with a real system providing frequency regulation. It requires as input a set of variables, i.e., the requested power to BESS ($P_{reg}(t)$), a feedback on the BESS SoC (SoC(t-1)), and the ambient temperature. It delivers as output the actual power delivered by the BESS ($P_{del}(t)$), its efficiency, and its SoC evolution (SoC(t)). It features a variable efficiency with respect to requested power and SoC. It features a capability curve that limits the maximum available power with respect to SoC (i.e., charge power is limited at very high SoC and discharge power at very low SoC). It features auxiliary systems to replicate the ones of a mid-tolarge-scale BESS (e.g., HVAC, fans, fire alarm, SCADA, etc.), with an electricity demand variable with ambient temperature and power requested to BESS [36]. The model offers a trade-off between accuracy and computational time. Thus, it suits a study that requires a set of simulations (i.e., tens of simulations) for a long period (i.e., a monthly period).



Fig. 4. A sample of the provision of aFRR (left) and RR (right), upward (orange) and downward (blue). The dashed lines depict the bid quantities, the shaded areas represent the awarded bids, the black line is the power requested in the real-time provision.

2.4. The balancing market modelling

A simplified BM model is implemented with the aim of defining if each bid is awarded. A maximum accepted price for upward reserve and a minimum accepted downward price are defined for each hour: these represent the marginal prices for the market. They are estimated based on a statistical analysis of the Italian ASM, performed on BM data for the period 2017-2019, to have prices for a business-as-usual situation (i.e., before Covid-19 and war in Ukraine). These marginal prices are the less convenient (from the system perspective) prices that are awarded on the market for a specific market session. The average marginal prices and standard deviations coming from the statistical analysis of BM are adopted to define a normal distribution of the hourly marginal awarded prices for each regulation, for both working days (Monday to Friday) and holidays (Saturdays, Sundays and bank holidays). Then, these outcomes are elaborated to feed the model with a random hourly value coming from the probability distribution for that hour of the day. The distribution of the marginal prices fed to the model is reported in Fig. 5. The probability distributions have been cut to avoid unrealistic (or unfeasible) prices: the upward prices cannot be lower than 60 €/MWh (to be equal or greater than DAM prices in 2017-2019), while downward prices cannot be lower than 0 €/MWh (no negative prices are assumed).



As mentioned before, the BESS is included in an energy district and aims to provide both self-consumption enhancement and ancillary services to the grid. The adopted strategy aims to first guarantee maximisation of self-consumption, then to bid the BESS' available power and energy margins on ASM. While the rationale and the details of this control strategy were first presented and are better illustrated in [41], we describe in the following how it is implemented in this study.

2.5.1. Available energy margins estimation

To define the available margins in a limited energy reservoir, we consider the energy content at the market gate closure time (GCT) and the estimated energy variation in the next relevant period. Fig. 6 highlights what is meant by the relevant period for available energy estimation, which depends on the market timing. From Fig. 6, we see there is a superposition between market sessions, thus the available energy for the next BM session is related to what has been awarded in the previous session, too. This general framework is used as the basis for the serial analysis. As previously indicated, the analyses will feature different distances to delivery and delivery times.



Estimating energy content at the end of the session reveals the esti-

Fig. 5. Outcomes of Balancing Market model relevant to the award/rejection rule: (a) probability distribution of maximum upward prices; (b) probability distribution of minimum downward prices.



Fig. 6. Schematic view of the balancing market session structure, highlighting the definition of distance to delivery, delivery time and relevant period adopted in this work.

mated available margins with respect to minimum and maximum SoC: these can be offered in the BM. The following elements are of interest when considering the BESS under study (energy flows are positive in case of battery discharging).

• The gap between SoC at GCT ($SoC(h - t_{DtD})$) and SoC_{min} is the margin for upward regulation (ΔE_{up}) at the market closure. Similarly, the gap between SoC_{max} and $SoC(h - t_{DtD})$ represents the energy content for downward (ΔE_{dn}).

$$\Delta E_{up} = \frac{SoC(h - t_{DtD}) - SoC_{min}}{100} * E_n \tag{2.2}$$

$$\Delta E_{dn} = \frac{SoC_{max} - SoC(h - t_{DtD})}{100} * E_n$$
(2.3)

• The estimated energy requirement for self-consumption (E_{SCest}) in the next relevant period (as highlighted in Fig. 6) is computed considering the energy district overall control logic illustrated in Paragraph 2.5.3, adopting two separate prediction models for the PV production and the load as detailed in paragraph 3.1. E_{SCest} can be either positive (consumption > production) or negative and is estimated as in (2.4).

$$\begin{cases} E_{SCest} = \sum_{i=1}^{t_{DD}+t_{del}} \frac{P_{sc_{-i}}}{\eta_{avgDis}} if \sum_{i=1}^{t_{DiD}+t_{del}} P_{sc_{-i}} \ge 0\\ E_{SCest} = \sum_{i=1}^{t_{DiD}+t_{del}} P_{sc_{-i}} * \eta_{avgCh} if \sum_{i=1}^{t_{DiD}+t_{del}} P_{sc_{-i}} < 0 \end{cases}$$
(2.4)

where η_{avgDis} and η_{avgCh} are constant values considering average discharging and charging BESS efficiencies.

• The estimated energy exchange for the provision of previously contracted market services (E_{ASMest}) must be computed. Indeed, the last part of the previous market session occurs after the GCT of the next session. The energy variation impacts the available margin for the next bid. E_{ASMest} is computed as in (2.5).

$$E_{ASMest} = \frac{P_{ASMup}(h - t_{DtD})^* r_{\underline{E}}^* t_{DtD}}{\eta_{avgDis}} - P_{ASMdn}(h - t_{DtD})^* r_{\underline{p}}^* t_{DtD}^* \eta_{avgCh}$$
(2.5)

where $P_{ASMup}(h - t_{DtD})$ and $P_{ASMdn}(h - t_{DtD})$ are the awarded power respectively for upward and downward service for the period between GCT and the start of the delivery time. $r_{\frac{E}{p}}$ indicates the hourly MWh requested per each awarded MW. While $r_{\frac{E}{p}}$ is 1 for RR, it is 0.34 for aFRR. This comes from a statistical analysis of the gross energy requested for the provision of aFRR (both upward and downward) in each period of 4

h of 2019 with respect to the awarded power: this means that for 1 MW awarded, the hourly energy provided is 0.34 MWh, with balanced upward and downward requests. It is worth noting that: in case of RR provision, either the first or the latter term (or both) is zero for each hour, since the award of bids is mutually exclusive (either no one or just one bid is awarded); in case of aFRR, awarded upward and downward power can be both nonzero (in particular, they must be equal for symmetric provision).

• The auxiliary system demand (E_{auxEst}) is the estimated auxiliary systems' energy demand for the hours between $[h - t_{DtD}, h + t_{del}]$, obtained by the BESS model [36].

Finally, the available upward and downward energy content are calculated as in (2.6) and (2.7), respectively.

$$E_{avUp} = max(0, \Delta E_{up} - E_{SCest} - E_{ASMest} - E_{auxEst})$$
(2.6)

$$E_{avDn} = max(0, \Delta E_{dn} + E_{SCest} + E_{ASMest} + E_{auxEst})$$
(2.7)

These equations ensure all factors affecting the available energy are considered. Null values are offered on the market in case the energy assessment for one of the two bids yields negative results, i.e., no margins are available. A schematic representation of the available energy estimation for a generic BESS + PV hybrid plant implemented in an energy district is given in Fig. 7. As can be seen, the magnitude of the available energy contents is likely to change due to time of day and previous outcomes on ASM. Indeed, the availability of upward energy (and consequently the maximum feasible upward bid) is usually larger during daytime and in case of previous downward calls (top right of Fig. 7). Oppositely, available energy for downward provision is larger during nighttime and following a previous upward call (bottom left).

 E_{avUp} and E_{avDn} are divided by t_{del} hours and by r_{E} to obtain the bid

quantity in kW. This is checked against the maximum power that can be dedicated to ancillary services provision (P_{maxASM}) as in (2.8) and (2.9). For the sake of the reliability of the provision, P_{maxASM} is 50% of P_n : this is a parameter obtained after a preliminary fine-tuning (this is a simplified hourly check that does not assure the reliability of provision in case of spikes, e.g., due to self-consumption). To avoid micro-bids, a minimum threshold (P_{minASM}) is foreseen, too. This is also to avoid compromising the BESS efficiency, which gets very low for small power output [36].

$$\begin{cases} P_{bidUp} = min\left(\frac{E_{avUp}}{t_{del} * r_{E}}, P_{maxASM}\right) if \frac{E_{avUp}}{t_{del} * r_{E}} > P_{minASM} \\ P_{bidUp} = 0 \ elsewhere \end{cases}$$
(2.8)



Fig. 7. Schematic representation of the available energy. At the gate closure time, the energy variation for self-consumption (SC), ASM participation and auxiliary demand within the end of the next market session is estimated and available energy is computed. Four cases are shown: a) daytime and upward call in the previous market session; b) daytime and downward call; c) nighttime and upward call; d) nighttime and downward call.

$$\begin{cases}
P_{bidDn} = min\left(\frac{E_{avDn}}{4}, P_{maxASM}\right) if \frac{E_{avDn}}{4} > P_{minASM} \\
P_{bidDn} = 0 elsewhere
\end{cases}$$
(2.9)

Where both P_{bidUp} and P_{bidDn} are absolute values representing the bids for upward and downward reserve, respectively.

If the provision is symmetric, the upward and downward bid quantities must be equal: in a conservative strategy, they are reduced to the minimum between the available quantities.

$$P_{bid}(t) = min(P_{bidDn}(t), P_{bidUp}(t))$$
(2.10)

2.5.2. Balancing market award/rejection rule

After bidding, the BM model returns the award/rejection of the bid. This is based on a purely economic criterion that is aimed to represent a possible simplified TSO's behavior for selecting resources.

- For upward reserve, the bid is awarded if the bid price (to receive) is lower or equal with respect to the maximum awarded price for that market hour.
- For downward reserve, the bid is awarded if the bid price (to pay) is higher or equal with respect to the minimum awarded price for that market hour.

Fig. 8 exemplifies the award/rejection rule for upward service provision. Bids are awarded (accepted, green dot) or rejected (not accepted, red dot) if the bid price (ϵ_{up}) is respectively lower or higher than the maximum accepted price for that hour (ϵ_{upMax}), considering marginal prices coming from BM model (paragraph 2.4).

As mentioned in paragraph 2.2 either or none of the bids can be awarded for aFRR. In case of RR, no double simultaneous award of downward and upward bids are possible. In case both prices are more convenient than the maximum/minimum, a criterion based on the spread is applied. If the spread between the upward offer and the DAM price is lower than the one between DAM price and downward offer, then only upward offer is awarded, and vice versa. The hourly awarded power on ASM ($P_{ASM}(t)$) is then provided.

2.5.3. The self-consumption logic in an energy district

The considered plant layout is a grid-connected smart energy district [49–51]. It considers an electric load that must be satisfied using local energy production or withdrawal from the grid. The generating assets are a combined heat and power (CHP) plant and a PV plant. A BESS is



Fig. 8. Award/rejection rule for upward service provision:

also present. The considered energy flows only focus on electricity, while heating needs are disregarded. The layout and the control strategy illustrated below aim to provide a general case study, whose outcomes could be immediately applicable to a large share of DERs and easily extended.

The Reference Case logic considers maximisation of energy selfconsumed. Priority is given to PV production, that is consumed as it is available (see eq. (2.11)). The CHP follows the residual electric load ($P_{residual}(t)$) as in eq. (2.12) (i.e., the heat is a by-product, and it is disregarded for the sake of the study).

$$P_{residual}(t) = P_{load}(t) - P_{pv}(t)$$
(2.11)

$$P_{CHP}(t) = \min(P_{nCHP}, \max(P_{tmCHP}, P_{residual}(t)))$$
(2.12)

where: $P_{CHP}(t)$ is the CHP setpoint also considering its nominal power (P_{nCHP}) and its technical minimum (P_{tmCHP}), i.e., the power below which the CHP cannot operate safely and should shut off; $P_{load}(t)$ is the load demand; $P_{pv}(t)$ is the PV generation. Therefore, the CHP is never shutting off. The BESS activates if the difference between the CHP setpoint and the residual load is not null.

G. Rancilio et al.

$$P_{SC}(t) = P_{residual}(t) - P_{CHP}(t)$$
(2.13)

where $P_{SC}(t)$ is the requested BESS power for self-consumption. Since the CHP is not shutting off, the $P_{SC}(t)$ can be:

- positive (discharge) in case the residual load is larger than the CHP nominal power;
- negative (charge) in case the residual load is smaller than the CHP technical minimum;
- null elsewhere.

PV and CHP detailed modelling is disregarded, and only the output power profiles are considered. For a better understanding of the self-consumption strategy, please refer to Fig. 10 and paragraph 3.

This priority-based control strategy is optimal in case the PV generation cost is lower than the CHP marginal cost, that is lower than the BESS levelized cost of cycling, that is lower than the spread between bill cost and injection value. The plant layout and the energy flows are proposed in Fig. 9. For the Reference Case just depicted, only the black dashed lines are of interest (no flexibility provision).

2.5.4. Power setpoint for multiservice provision

The overall power requested to the BESS $(P_{req}(t))$ is computed as follows.

$$P_{req}(t) = P_{SC}(t) + P_{ASM}(t) + P_{aux}(t)$$
(2.14)

This equation includes the self-consumption enhancement in the energy district ($P_{SC}(t)$) computed as described in paragraph 2.5.3, the provision of ancillary services ($P_{ASM}(t)$) as per the BM award/rejection rule in paragraph 2.5.2, and the feeding of auxiliary systems electric demand ($P_{aux}(t)$) estimated via the auxiliary system model described in paragraph 0. $P_{req}(t)$ is fed to the model as a 1-min setpoint. The $P_{req}(t)$ is checked against the BESS model. The actual power that can be delivered $P_{del}(t)$ could be equal or different from $P_{req}(t)$. Differences can occur if SoC limits are hit, or capability curve limits the power. In case delivered power is different from requested, imbalances occur. If ancillary services request a power provision from/to the grid, this is not accounted as an exchanged power generating imbalance. The power exchanged with the grid ($P_{exch}(t)$) is therefore the difference between request and delivery, net of the ancillary services provision.



Fig. 9. Plant layout with energy (black) and flexibility flows (green).

$$P_{exch}(t) = \left(P_{req}(t) - P_{ASM}(t)\right) - P_{del}(t)$$
(2.15)

It is positive in case of withdrawal, and negative in case of injection. The adopted Multiservice strategy is not optimized, but it is considered as a possible standard routine for BESS aimed to revenue stacking in the context of an energy district (or, in general, of DERs).

2.6. Assessing the optimal ASM arrangements for opening to DERs

As already pointed out in section 1, the scope of work is to define a possible correlation between each market parameter and the BESS performance. If correlations are found, an optimal range of parameters should be highlighted. The performance should be evaluated from different perspectives. The adopted Key Performance Indicators (KPIs) are listed in the following. The performance of each ASM scheme is checked against a reference case (REF) featuring the provision of selfconsumption only, as described in paragraph 2.5.3, without participation to ASM. The energy that can be provided for both the services delivered by the Multiservice strategy is considered. To compare the situation with and without the participation to ASM, the self-consumed energy (E_{sc}) in MWh in each case is evaluated with respect to the E_{sc} in the REF case. This way, we assess if the ASM participation improves the results with respect to a standard BESS control strategy, only devoted to self-consumption. This is considered coherent with the analysis, that aims to return the optimal ASM design for maximising the sum of system's and operators' benefits with respect to no participation. It is worth noting that self-consumption creates not only value for the energy district in terms of avoided costs, but also for the network in terms of reduction of exchanged energy and flows in local distribution network [52-54].

To assess the performance of self-consumption, the total energy requested for self-consumption in MWh is estimated (E_{SCreq}).

$$E_{SCreq} = \frac{t_s}{3600} \sum |P_{SC}(t)|$$
(2.16)

Therefore, the E_{screq} is the absolute integral of the power requested for self-consumption and t_s is the sampling time in seconds (in this case, 60 s since the simulation sampling rate is 1 min). The power nonprovided for self-consumption ($P_{SC,NP}(t)$) and as a consequence the total non-provided energy ($E_{SC,NP}$) are considered as follows.

$$P_{SC,NP}(t) = \min(P_{SC}(t), P_{exch}(t))$$
(2.17)

$$E_{SC,NP} = \sum |P_{SC,NP}(t)| \tag{2.18}$$

And the actual energy delivered for self-consumption is therefore:

$$E_{SC} = E_{SCreq} - E_{SC,NP} \tag{2.19}$$

In addition, a self-consumption provision ratio (SC%) is estimated as follows.

$$SC\% = \frac{E_{SC}}{E_{SCreq}}$$
(2.20)

For what concerns the performance on ASM, both the total energy provided (E_{ASM}) and its reliability are of interest. Reliability is complementary to the nonprovision (NP). Nonprovided power in each instant ($P_{NP}(t)$) is estimated as follows.

$$\begin{cases} P_{NP}(t) = P_{req}(t) - P_{del}(t) - P_{SC,NP}(t) if \frac{|P_{del}(t) + P_{SC,NP}(t)|}{P_{req}(t)} > 5\% and P_{ASM}(t) \neq 0\\ P_{NP}(t) = 0 elsewhere \end{cases}$$

$$(2.21)$$

where a 5% of inaccuracy in the provision is tolerated and does not entail penalties [55]. NP (in %) is then obtained as the integral of $P_{\text{NP}}(t)$ over the E_{ASM} , as follows.

$$NP = \frac{\sum |P_{NP}(t)|}{\sum |P_{ASM}(t)|}$$
(2.22)

Both BESS and system operator aim to enhance the flexibility made available by the resource. The energy sold on ASM (EASM) is estimated as a KPI as the gross total of awarded energy for downward and upward provision. We consider energy since the remuneration on the Italian market is energy-based (€/MWh). Nevertheless, given the importance it has for the system operator, the power available for providing flexibility should be considered, too. To do so, the average power that is bid on the ASM in both upward and downward direction (PbidASMup, PbidASMdn) is estimated. It is worth noting that the Multiservice strategy considers the ASM participation as a secondary application. The bid quantity, as deeply illustrated in paragraph 2.5.1, is the total available power band, considering the battery limitations and the energy flows devoted to the provision of the main application, that has the priority. Therefore, the P_{bidASM} can be considered the flexibility that can be made available by the DERs, constrained by the market arrangements, and causing no further effort to the energy district. Hence, it is a measure of the available flexibility, given an ASM arrangement. Being able to increase the offered MW on ASM with a fixed installed capacity of resources is something to be pursued by an effective market design [56].

The reliability of provision is a KPI for the system. The provided flexibility should be delivered by keeping as small as possible the difference between the requested energy and the activated energy. The reliability of provision (*Rel*) is estimated as complementary to 1 of the NP, as follows.

$$Rel = 1 - NP \tag{2.23}$$

A higher reliability could foster the penetration of the DERs in ASM, since these resources would be considered trustable as an alternative to reliable conventional generators, traditionally involved in electricity balancing.

The BESS efficiency is a KPI for the operator since BESS variable opex increases if efficiency decreases. In this study, the BESS efficiency ($\eta_{BESSavg}$) is estimated as the average roundtrip efficiency during 1-month operation.

The ASM participation has the goal of improving the economics of BESS operation. To have a general estimation of the BESS net revenues (i.e., providing conclusions that are not bound to a specific regulatory and market framework), the following statements can be proposed.

- The economic attractiveness of the project increases if the selfconsumed energy increases. Indeed, self-consumption is generally correlated with avoided costs, but this depends on the entity of grid costs in electricity tariff, that can change [57]. Thus, the high E_{SC} is considered in this study as a minor driver for the economic attractiveness of investments in BESS.
- The flexibility sold on the market is directly correlated with the economic attractiveness of the investment in BESS. As shown in [41], indeed, ASM participation is one of the most remunerative applications for BESS. The E_{ASM} is therefore considered a main driver for improving the economics.
- Non-performance (NP) is generally associated with penalties (NPP) that shrink the net revenues on the ASM. Below a certain threshold, the lack of reliability can even lead to exclusion from the provision (or from the remuneration). Therefore, high reliability is considered a main driver for economics.
- BESS inefficiencies are responsible for additional requested charging energy, thus more opex, but BESS is generally highly efficient. Therefore, BESS efficiency is a minor driver of the economic attractiveness of investment in BESS.

A semi-quantitative economic KPI is included in the results of the study as a combination of the drivers listed above. The equation to compute the overall economic evaluation of each ASM arrangement (EE_i) is the following.

$$EE_i = \sum_j^{\gamma} D_{ij} * w_j \tag{2.24}$$

where D_{ij} is a score from 0 to 3 and w_j is a weighting factor based on what just illustrated. Therefore, EE_i returns the sum of the scores of the four drivers for each simulated case *i*. Table 1 reports the values, the weights, and the sources used for defining the scores.

The described KPIs, all together, return a comprehensive evaluation of the suitability of an ASM arrangement for the inclusion of DERs.

3. The case study: An energy district in the tertiary sector that sells flexibility on balancing markets

A wide set of 1-month simulations of BESS operation is performed considering the previously illustrated layout. The difference between each simulation is the ASM arrangement, varying as illustrated in Paragraph 2.1. To characterise the case study with real-world data, the Politecnico di Milano's "Leonardo" Campus is considered. It is the headquarter of the university, characterized by 21 main buildings and >20 thousand students and researchers hosted every day, with classrooms, labs, offices, and shops. Data on this Campus are recorded and made available by the "Commissione Energia", a technical group based in Politecnico di Milano [59].

The energy district of Leonardo Campus can be described as follows [49]:

- a total load of 4000 kW,
- a CHP plant of 2750 kW of nominal power, with a technical minimum of 1375 kW.

A PV generator of 1600 kWp is added in the case study, as per the projects in Politecnico di Milano [60]. In the case study, the CHP operates following the electric load. In addition to this layout, a BESS and the power grid must be considered, as seen in Fig. 9. The BESS data are given in Table 2 and it is modelled as already illustrated. The BESS power is selected to be able to satisfy the peak demand. The sizing in energy is coherent with the maximum daily energy output by the PV obtained with PVGIS [61].

The reference operating strategy for the BESS is to cope with the residual self-consumption requested by the plant, as described in Paragraph 0. The power flows in some standard weekdays and holidays are shown in Fig. 10. The PV generates power during the daytime (in yellow). The electric load (in red) is higher during the daytime but is not null during the nighttime (around 2000 kW). To the left part of the figure, two weekdays are depicted (peak load around 3500 kW). To the right part, Saturday and Sunday are shown (similar pattern, but peak slightly above 2000 kW). The PV production is primarily serving the load. For limited periods, it is injected towards the grid, e.g., at the weekends. The CHP is always on, ranging between the technical minimum and the nominal power, to cope with the residual load (i.e., the difference between the red and yellow curve). Since the CHP provides heating, cooling, and other services to the Campus [62], it is never shut off. Therefore, if the residual load is lower than the technical minimum, the district features an excess in production (e.g. in the morning on weekdays and in weekends). The BESS operates to further enhance selfconsumption (see green and purple lines): it is requested to charge when production exceeds the load, it is requested to discharge when generation lacks.

3.1. Energy and power profiles and forecasts

The load and PV profiles of the district for the considered month (June 2021) are presented in Fig. 11. They are retrieved from 15-min average data.

Table 1

Scores and weights for each driver.

#	Drivers	0	1	2	3	Weight (w _j)	Source
1	SC%	<70%	70–80%	80–90%	90-100%	1	[58]
2	E _{ASM} (MWh)	<10	10-50	50-200	>200	2	[22]
3	NP	<90%	90-95%	95–98%	98–100%	2	[55]
4	$\eta_{BESSavg}$	<70%	70–80%	80-85%	>85%	1	[38]

Table 2	
Energy district's BESS data.	

Key	Value	Unit
Technology Nominal neuror	Li NMC	N/TA 7
Nominal power	4 10	MWh

As well as actual power profiles, forecast profiles must be fed to the model for the estimation of the available bands. It is performed with a data-driven model based on classification trees. Two different models are trained for PV and load prediction. The main metadata of the models are given in Table 3, they refer to the prediction of the quantities per unit with respect to their nominal power (thus, they are given in %). It is worth noting that the adopted models are standard, for having generalizable results.

The actual and forecasted power profile are given as input to a Simulink tool, where the BESS model is implemented as well as the





Fig. 10. A sample of the considered power flows in the energy district.

Fig. 11. The load and PV profile for the considered 30 days.

Table 3

Metadata of the PV and load prediction models.

Key	PV model	Load model
Method	Bagged Tree Hour of the day, PV	Least-square Boosted Tree Hour of the day, day type (i.e.,
Features	production at h – 24 (i.e., the day-ahead).	weekday, holiday), Load at h – 168 (i.e., the week ahead).
Root Mean		
Squared Error (RMSE)	5.2%	3.6%
Mean Absolute Error (MAE)	2.9%	2.8%

model of the district as per the equations presented in Paragraph 2.5.3.

3.2. Setup of the simulations

The simulated case studies differ based on:

- the considered control strategy, that is self-consumption (SC) only in the reference (REF) case (see paragraph 2.5.3), while it is Multiservice in all other cases (see paragraph 0);
- the considered ancillary service and market arrangements, where the previously introduced parameters vary in a range to return a sensitivity analysis of BESS performance under different ASM arrangements.

The study features 25 simulations. The summary of the performed simulations is provided in Table 4. Both aFRR and RR are tested both symmetrically and asymmetrically. The time definition (TD) ranges from 24 h to 1 h. The Distance to Delivery (DtD) ranges from 24 h to zero

Table 4

Layout of the simulation campaign.

Case	Service	Symmetry? [Y/N]	TD [h]	DtD [h]
REF	_	-	-	-
1	aFRR	Y	4	1
2	aFRR	N	4	1
3	RR	Y	4	1
4	RR	N	4	1
5	aFRR	Y	24	1
6	aFRR	Y	8	1
7	aFRR	Y	2	1
8	aFRR	Y	1	1
9	aFRR	N	24	1
10	aFRR	N	8	1
11	aFRR	N	2	1
12	aFRR	N	1	1
13	RR	N	24	1
14	RR	N	8	1
15	RR	N	2	1
16	RR	N	1	1
17	aFRR	N	4	24
18	aFRR	N	4	4
19	aFRR	N	4	0.5
20	aFRR	N	4	0
21	RR	N	4	24
22	RR	N	4	4
23	RR	N	4	0.5
24	RR	N	4	0

(immediate delivery). The several simulations performed are necessary to provide a modular analysis. Between a simulation and the next one, only one parameter (i.e., service, symmetry, TD, or DtD) changes: this allows to detect the specific effect of each parameter on the KPIs. The considered ranges are coherent with the provision of ancillary services in several analyzed markets: indeed, markets are generally characterized by either asymmetric or symmetric procurement, by daily to hourly products, contracted from the day-ahead to real time.

4. Results and discussion

The summary of the resulting KPI for the 25 simulations is given in Fig. 12.

One first consideration is on symmetric RR. In the case of the RR provision (as per the proposed rules), it is meaningless to test the symmetric service: it would result in a simultaneous provision of two constant setpoints of the same magnitude, a positive and a negative one. This is tested in Case 3, where the obtained results are the same as in Case REF since there is no provision on ASM: the results of this test are not proposed in Fig. 12 to avoid confusion, and no more tests with symmetric RR are proposed.



Fig. 12. Main results of the simulation campaign on the ASM arrangements.

4.1. Energy self-consumed

For what concerns the self-consumed energy (E_{SC}), most of the cases show a higher value with respect to REF case. The Multiservice strategy increases self-consumption with respect to the standard control strategy focused on self-consumption only, since the offered margins on ASM work as SoC management. As proposed in Fig. 13 (from case REF = 0, 2, 4), the provision of self-consumption largely requests the BESS to inject power, while the battery charge rarely occurs (e.g., during weekend daytime). Thus, the battery suffers self-discharge due to the auxiliary systems and depletes its energy content. The addition of aFRR and even more of RR allows to increase the battery cycling, avoiding the depletion of SoC and increasing the availability to self-consume.

The increase of self-consumption with respect to REF case always occurs except for Case 1, and 5–8. These Cases feature symmetric provision of aFRR. It does not allow an effective implicit SoC management: the combined effect of the aleatory behavior of aFRR and the reduced bands according to eq. (2.10) do not often support the BESS to restore SoC towards the target. While providing an asymmetric service, the self-consumption is generally high: it is higher for RR than for aFRR. In general, neither TD nor DtD have a strong influence on the self-consumption: self-consumption slightly decreases for small TD, since a larger quantity of energy is offered on the ASM and this leads to competition between the provision of the two services, in the end reducing both the self-consumption and the reliability on the ASM.

4.2. Energy sold on ASM and flexibility

In this paragraph, we analyze both the energy sold on ASM and the flexibility made available on it.

The E_{ASM} is the gross amount of energy awarded on the market, for both upward and downward provision. As presented in Fig. 12, the offered energy largely changes. The simulation 1–4 highlight that a small amount of energy is sold in case of provision of symmetric services: symmetric aFRR only sells 16 MWh of monthly flexibility. The E_{ASM} increases to 148 MWh in case of asymmetric aFRR, and a further step up to 273 MWh sold occurs in case of asymmetric RR.

There is a strong inverse correlation between the energy traded on ASM and the time definition of products. Indeed, a shorter delivery time allows a limited energy reservoir to increase the bid power. An hourly time definition, with respect to a time definition of 4 h, allows to trade from 1.8 (for aFRR, simulation 10 vs 12) to 5.7 times (for RR, simulation 14 vs 16) the EASM. The energy sold for RR is larger with respect to a coherent provision of aFRR by 1.5 to 4.6 times (comparing simulations 9-12 to 13-16). This is not because the RR is more energy-intensive than aFRR (see eq. (2.5), the aFRR bids are larger to balance the lower energy request) but because of the aleatory behavior of aFRR. Indeed, as can be seen by Fig. 14, comparing SoC distribution of aFRR (left) and RR (right) provision, RR better spans the SoC range. The RR allows to better exploit the energy content of the BESS: e.g., the SoC gets high after delivery of services, thus the P_{bid} is high for upward service, and if awarded, it gets close to minimum SoC, thus allowing offering a large amount of downward reserve for next session and increasing the daily cycles. This is a virtuous circle, since the large energy provision increases the BESS efficiency, thus leading to a SoC distribution considerably shifted towards high SoC.

As said, a requirement for an ASM arrangement to be defined efficient is the possibility to exploit a large amount of flexibility even by a limited installed capacity of resources. The KPI to evaluate this is the average power (P_{bidASM}) that is offered on the market in the simulated month (P_{bidASM}). As per the Multiservice Strategy, this power is represented by the available margin left after providing the primary service (self-consumption, in this case). Focusing on the asymmetric provision, Fig. 15 shows the trend of bid power for aFRR (in red) and RR (in blue). The awarded power depends on the SoC. Nonetheless, the services show differences: at the same SoC, the aFRR can bid a sharply larger power.



Fig. 13. Fifteen days of SoC evolution during service provision for self-consumption only (REF case in black), for aFRR (Case 2 in red) and RR (Case 4 in blue).



Fig. 14. SoC distribution for aFRR (Case 11, left part) and RR (Case 15, right part).



Fig. 15. Bid power (top chart) and SoC (bottom chart) profiles for aFRR (Case 2) and RR (Case 4).

This is because of the smaller energy requirements of aFRR with respect to RR, considered by the Multiservice strategy as per eq. (2.8). In case of a time definition of 4 h, the $P_{bidASMup}$ for aFRR is 1.6 MW, while the $P_{bidASMdn}$ is 3.8 MW (Case 2). For RR, the values are respectively 1.0 and 1.2 MW (Case 4). Reducing the time definition to 1 h, the pair ($P_{bidASMup}$, $P_{bidASMdn}$) for aFRR increases to (2.3, 4.0) MW, and (3.0, 3.5) MW are the values for RR (Case 12 vs 16). Therefore, the bid power is inversely proportional to time definition, too. For aFRR it is generally unbalanced towards downward power. This is because of the lower real-time power flows and the consequently lower efficiencies (and larger weight of auxiliaries) bringing down the SoC. It is worth noting that for low time definitions (e.g., 1 h), in both cases we have 2.4–4.0 MW of flexibility averagely available on the ASM for a BESS of 4 MW of nominal power: this means that 60 to 100% of the installed power can be regularly devoted to flexibility.

4.3. BESS efficiency

The simulations revealed a wide range of average roundtrip BESS efficiencies. Efficiency is very low (down to 45%) for small E_{ASM} . In the REF case, for instance, the minimum efficiency is recorded since the battery is often subject to discharge for feeding the auxiliary systems' demand (see Fig. 13), hence increasing the auxiliaries' share on losses. The efficiency rises to 85% in case of massive participation to ASM. We can conclude that there is interest in exploiting the BESS as much as possible to increase the operating system efficiencies up to doubling them.

4.4. Reliability of ancillary services provision

The reliability of the provided services is generally high: reliability is above 90% for 20 out of 22 relevant Cases (see bottom chart of Fig. 12). This is given by the effectiveness of the Multiservice Strategy, only bidding what can be provided. In addition, this proves the trustworthiness of BESS and DERs on ASM, thus backing their larger exploitation. The reliability is above 97% for large time definitions (4 to 24 h) and relatively small distances to delivery (<24 h for RR, <4 h for aFRR). This is due to the significantly lower E_{ASM} in these cases. For small time definitions, the traded energy increases, leading sometimes to the impossibility of providing all the energy, e.g., due to SoC saturation. This reduces reliability to smaller, yet acceptable values: around 90% for aFRR and 95% for RR in case of 1 h of time definition.

The effect of the distance to delivery is clearer on the aFRR, since its behavior is aleatory: the provision of aFRR itself can lead SoC to diverge between the commitment (the market closure) and the delivery time, thus leading to unreliability. This is a drawback, for instance, for products contracted on the day-ahead [63].

4.5. Comprehensive economic evaluation and optimal ASM arrangements

To achieve generalizable results, the economic analysis evaluates the set of previously described KPIs that are considered drivers of the economic attractiveness of the investment. The main drivers are the energy traded on the ASM and the reliability in providing the ancillary services. The economic evaluation (EE_i as computed in eq. (2.24)) resulting when considering all KPIs is presented in Fig. 16. In the figure, the colours

Case	Service	Symmetry? (Y/N)	TD (h)	DtD (h)	Economic Evaluation (EE)
REF	-	-	-	-	
1	aFRR	Y	4	1	
2	aFRR	N	4	1	
3	RR	Y	4	1	
4	RR	N	4	1	
5	aFRR	Y	24	1	
6	aFRR	Y	8	1	
7	aFRR	Y	2	1	
8	aFRR	Y	1	1	
9	aFRR	N	24	1	
10	aFRR	N	8	1	
11	aFRR	N	2	1	
12	aFRR	N	1	1	
13	RR	N	24	1	
14	RR	N	8	1	
15	RR	N	2	1	
16	RR	N	1	1	
17	aFRR	N	4	24	
18	aFRR	N	4	4	
19	aFRR	N	4	0.5	
20	aFRR	N	4	0	
21	RR	N	4	24	
22	RR	N	4	4	
23	RR	N	4	0.5	
24	RR	N	4	0	
Poor		Fair	Good		Excellent
(4th	quartile)	(3rd quartile)	(2nd	quartile)	(1st quartile)

Fig. 16. Economic evaluation of each case.

represent the ranking of each simulation: the first quartile is in bright green, presenting the simulations that have an overall economic evaluation included in the first quartile of results, and so on towards red colour, representing the quartile with the lowest economic evaluations.

A poor to fair economic interest is recognized for REF case and for revenue stacking based on symmetric provision of ancillary services. More positive outcomes are shown for asymmetric provision, in particular of RR. AFRR hardly gets an excellent evaluation (1st quartile): a good outcome is achieved for 4–8 h of time definition and for distance to delivery equal or lower than 1 h. The asymmetric RR provision returns excellent economics for 2–8 h of time definition and for distance to delivery lower than 4 h. Apparently, a lower DtD is necessary for services whose energy demand is characterized by an aleatory nature (e.g., depending on frequency deviation or FRCE). In addition, a mid-to-high time window for delivery (TD) is preferable, otherwise the reliability could decrease. Indeed, with a lower TD, a larger power is bid and can lead to strong energy demand in brief time, depleting or saturating the energy content of BESS. The DtD has instead less impact on a service based on static setpoint (as RR).

This analysis helps identifying an optimal range of parameters for standard balancing products. They are shown in Fig. 17. They are coherent with the adopted ranges in the Cases achieving the highest grade of evaluation in Paragraph 4.5.

The results for aFRR can be generalized, up to a certain extent, to all the services showing an aleatory behavior and a low energy intensity. For instance, frequency response services (including Fast Frequency Response) and the services responding to the FRCE. They need to be contracted closer to real time. They can be provided for a generally long period, since they do not require an intense energy provision.

For what concerns the RR (bottom part of Fig. 17), the results can be generalized to the provision of RR, congestion management, balancing and other slow dynamics services. The ranges are larger in this case, and the achieved results are better. Nonetheless, BESS face larger



Fig. 17. Optimal ranges for the provision of a) aFRR (top chart) and b) RR (bottom chart).

competition when providing services characterized by a static setpoint or by slow dynamics. Indeed, these can be provided by a larger turnout of resources, not only inverter-based systems [64].

Concerning distance to delivery, the economic interest towards services traded daily (e.g., a daily auction that occurs the day ahead the delivery) is lower than for products traded just some hours ahead of the delivery. Indeed, if you increase the distance between GCT and delivery time, the compatibility with a revenue stacking vision (that is among the pillars of this study) decreases. As seen in Fig. 12, simulations 17 and 21 show both lower self-consumed energy and reliability on ASM. Logics can be different while deploying BESS integrated in other plants or standalone.

It is worth noting that, within the study, a sensitivity analysis has been performed on the BESS sizing, to check the results with an E/P ranging from 1.5 to 3. This range comes from a previous study, showing that in a revenue stacking framework, with BESS smaller than 10 MW, internal rate of returns larger than 5% are guaranteed for BESS of duration \leq 3 h [65]. It resulted that the provided outcomes on the optimal BESS arrangement are generally independent from E/P within this limited range.

In conclusion, it should be highlighted that the weights proposed in Table 1 quantitatively influence the results: ASM related KPIs have a large weighting factor since we assume that, today and in the future, BESS will be mostly dedicated to ancillary services provision, where larger revenues can be achieved. Additionally, different pricing mechanisms (e.g., capacity payment in ϵ /MW/period instead of ϵ /MWh), could alter the results quantitatively. Qualitatively, the analysis is considered robust.

4.6. Application to national frameworks

To better address the obtained results and gather some takeaways, the evaluation of real-world examples is provided (see Table 5). Describing the real-world experiences aims to highlight that the proposed evolutions are subject of actual regulatory reviews. It can be useful to understand whether the market reviews head towards a better symbiosis of BESS (and in general DERs) and flexibility markets or not.

Since 2021 the Italian BM changed from a setting with six 4-h market sessions a day, to hourly market sessions. Therefore, the provision of RR in Italian BM was previously represented by Case 4 of this study (see Table 4), while it is now performed as in Case 16. Considering the findings of this work, the expected performance of the energy district change as follows. The offered flexibility and, consequently, the energy traded on the ASM by each BESS is expected to grow sharply. The management of behind-the-meter services could remain the same, as well as the BESS operational efficiency. The reliability of the provision could slightly decrease, maintaining high values (e.g., 95%). Therefore, the evolution is in the direction of further including DERs in the electricity balancing and increasing the flexibility that can be provided by each resource.

Table	5			
Model	application	to	national	frameworks

National framework	Service	Previously represented by	Now represented by	Model-based prediction of improvements
Italy	mFRR and RR	Case 4	Case 16	A sharp increase in the volumes of flexibility provided by each BESS.
Italy	aFRR	Case 1	Case 2	Both energy consumption and provided flexibility drastically enhanced.
Germany	FCR	Case 5	Case 1	The study does not highlight major improvements.

For what concerns the provision of aFRR, a pilot project for its asymmetric provision has been presented and activated after 2020 [66]: its final implementation is expected in 2025 (with the issue of the framework document "TIDE" [67]). This change would modify the way aFRR is provided from the one depicted in Case 1 to Case 2. This would drastically improve the techno-economic performance of BESS, increasing the energy traded on ASM by even 7 times – with no expenses on the reliability (see Fig. 12) – and enhancing the self-consumption rates. The economic evaluation would benefit as well, switching a poor evaluation into a good one (see Case 1 vs Cases 2 in Fig. 16).

The study clearly shows a performance decrease in case the gate closure time is too far from the delivery time. Considering the German FCR, the gate closure time is on the day-ahead. The time definition is 4 h since 2019 and the provision is symmetric. Before, time definition was daily. There is not a dedicated Case representing this provision in our study, nonetheless, a symmetric provision of an aleatory service for 24 h is described in Case 5, while 4-h provision is proposed in Case 1 (DtD is always 1 h, which is not the case for Germany). Passing from Case 5 to Case 1 shows no substantial improvement in the overall performance. In real-world, results show that FCR prices did not decrease after the introduction of this new time definition [68]. Furthermore, the decrease in distance to delivery can be of utmost importance in the case of an aleatory service. Cases 17 to 20 support this statement: the reliability increases by 15%, as well as the self-consumption (+21%) and the overall economic evaluation (from "poor" to "good").

The same rationale can be applied to different standard balancing products. The analysis can be extended to evaluate different conditions (e.g., weekly auctions, steeper provision) given the flexibility and low computational effort of the adopted BESS model.

5. Conclusions

In this work, a study on the regulatory barriers for BESS in the provision of ancillary services is presented. The BESS is integrated in a simplified energy district based on the Politecnico di Milano's Leonardo Campus grid. It is selected to be general and to have results that could be applied to a wide range of frameworks. The BESS is providing selfconsumption and ancillary services adopting a Multiservice strategy, comprising self-consumption enhancement and ancillary services provision.

The provision of services by BESS shows generally high reliability in a wide span of market arrangements. A larger amount of energy traded on the ASM is not in conflict with the provision of self-consumption: on the contrary, the latter improves as well thanks to synergies exploited by the developed control and bidding strategies. These outcomes strongly support a gradual yet fast penetration of DERs in electricity balancing and in ASM participation. Fine-tuning the parameters allows to retrieve general principles. The asymmetric provision of services (i.e., procuring different downward and upward quantities from a resource) drastically improves the performance, the reliability, and the economic attractiveness of flexibility provision. The different nature of services influence the performance. A static service (i.e., constant power) allows more reliable provision and larger economic benefits. Oppositely, it raises competition since more resources can provide it with high precision. A service requesting to follow an aleatory and dynamic signal slightly reduces reliability and sharply shrinks the traded energy. Yet, it is less energy-intensive, therefore more flexibility (in terms of power) can be offered with the same installed capacity. Eventually, a fast response service decreases the competition electing BESS as the preferential provider. Energy-intensive services benefit of lower time definitions: BESS can offer larger flexibility since it is less constrained by the energy requirements, and the techno-economic performance improves. Aleatory services benefit also of a gate closure closer to delivery time: it is difficult to forecast the available flexibility on a long-term horizon.

The findings and methodology could be applied to optimize a future change in the Grid Codes on balancing services, supporting for instance the decision-making when creating a new service or reforming a standard product, but also to assess a lack of efficiency or a worsening of the techno-economic situation downstream of a reform. To show this latter aspect, the analysis has been applied to recent changes in real-world markets, such as Italian and German frameworks. The model proposes an estimate of the evolution occurred in German FCR that is in line with real-world results. The model estimated no major improvement in the economic evaluation of BESS providing FCR in a 4-h window with respect to daily window; indeed, the real-world prices have not decreased in the period 2020–2021 when the redesign occurred.

The indications and data gathered in this work could be of use for stakeholders (e.g., regulators, policymakers, transmission system operators) aimed to abate the barriers of ASM for an increasingly competitive future market.

CRediT authorship contribution statement

Giuliano Rancilio: Writing – review & editing, Writing – original draft, Visualization, Methodology, Data curation, Conceptualization. Filippo Bovera: Writing – review & editing, Methodology, Data curation. Matteo Spiller: Software, Methodology. Marco Merlo: Writing – review & editing, Project administration, Methodology, Formal analysis, Conceptualization. Maurizio Delfanti: Supervision.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

Data will be made available on request.

Acknowledgements

This study was partially carried out within the NEST - Network 4 Energy Sustainable Transition (D.D. n. 1561, 1.10.2022, PE0000021) and received funding under the Piano Nazionale di Ripresa e Resilienza (PNRR), Mission 4 Component 2 Investment 1.3, funded from the European Union - NextGenerationEU. This manuscript reflects only the authors' views and opinions, neither the European Union nor the European Commission can be considered responsible for them.

References

- [1] IEA. World energy outlook 2022. Paris; 2022.
- [2] National Grid ESO. Winter balancing costs review. London; 2023.
- [3] Blakers A, Lu B, Stocks M. 100% renewable electricity in Australia. Energy Aug. 2017;133:471–82. https://doi.org/10.1016/J.ENERGY.2017.05.168.
- [4] ENTSO-E. Developing balancing systems to facilitate the achievement of renewable energy goals. In: Brussels; 2011. Accessed: Jul. 31, 2023. [Online]. Available: www.entsoe.eu.
- [5] Brown TW, Bischof-Niemz T, Blok K, Breyer C, Lund H, Mathiesen BV. Response to 'Burden of proof: A comprehensive review of the feasibility of 100% renewableelectricity systems. Renew Sustain Energy Rev Sep. 2018;92:834–47. https://doi. org/10.1016/J.RSER.2018.04.113.
- [6] Blakers A, Stocks M, Lu B, Cheng C. The observed cost of high penetration solar and wind electricity. Energy Oct. 2021;233:121150. https://doi.org/10.1016/J. ENERGY.2021.121150.
- [7] Hirth L, Ziegenhagen I. Balancing power and variable renewables: three links. Renew Sustain Energy Rev Oct. 2015;50:1035–51. https://doi.org/10.1016/J. RSER.2015.04.180.
- [8] Gailani A, Crosbie T, Al-Greer M, Short M, Dawood N. On the Role of Regulatory Policy on the Business Case for Energy Storage in Both EU and UK Energy Systems: Barriers and Enablers. Energies 2020;13(5):1080. Mar. 2020, https://doi.org /10.3390/EN13051080.
- [9] Poplavskaya K, De Vries L. Distributed energy resources and the organized balancing market: A symbiosis yet? Case of three European balancing markets. Energy Policy 2019;126:264–76. https://doi.org/10.1016/j.enpol.2018.11.009.

- [10] Rancilio G, Rossi A, Falabretti D, Galliani A, Merlo M. Ancillary services markets in europe: Evolution and regulatory trade-offs. Renew Sustain Energy Rev 2022;154: 111850. https://doi.org/10.1016/j.rser.2021.111850.
- [11] Castagneto Gissey G, Dodds PE, Radcliffe J. Market and regulatory barriers to electrical energy storage innovation. Renew Sustain Energy Rev Feb. 2018;82: 781–90. https://doi.org/10.1016/J.RSER.2017.09.079.
- [12] IRENA. Electricity storage and renewables: Costs and markets to 2030. Abu Dhabi; 2017.
- [13] El Chami N. Europe's energy storage transformation. PV Tech Power. Vol 24 :15-22. 2020. Accessed: Sep. 08. [Online]. Available: https://www.energy-storage. news/europes-energy-storage-transformation/.
- [14] Cappers P, MacDonald J, Goldman C, Ma O. An assessment of market and policy barriers for demand response providing ancillary services in U.S. electricity markets. Energy Policy Nov. 2013;62:1031–9. https://doi.org/10.1016/J. ENPOL.2013.08.003.
- [15] Borne O, Korte K, Perez Y, Petit M, Purkus A. Barriers to entry in frequencyregulation services markets: Review of the status quo and options for improvements. Renew Sustain Energy Rev 2018;81:605–14. https://doi.org/ 10.1016/j.rser.2017.08.052.
- [16] Schittekatte T, Meeus L, Jamasb T, Llorca M. Regulatory experimentation in energy: Three pioneer countries and lessons for the green transition. Energy Policy 2021;156:112382. https://doi.org/10.1016/j.enpol.2021.112382.
- [17] Kryonidis GC, et al. Ancillary services in active distribution networks: a review of technological trends from operational and online analysis perspective. Renew Sustain Energy Rev Sep. 2021;147:111198. https://doi.org/10.1016/J. RSER.2021.111198.
- [18] Guggilam SS, Zhao C, Dall'Anese E, Chen YC, Dhople SV. Primary frequency response with aggregated DERs. Proceed Am Control Conf Jun. 2017:3386–93. https://doi.org/10.23919/ACC.2017.7963470.
- [19] Newell SA, Carroll R, Ruiz P, Gorman W. Cost-benefit analysis of ERCOT's future ancillary services (FAS) proposal. 2015.
- [20] Ocker F, Ehrhart KM, Belica M. Harmonization of the European balancing power auction: a game-theoretical and empirical investigation. Energy Econ Jun. 2018; 73:194–211. https://doi.org/10.1016/J.ENECO.2018.05.003.
- [21] Poplavskaya K, Lago J, de Vries L. Effect of market design on strategic bidding behavior: model-based analysis of European electricity balancing markets. Appl Energy Jul. 2020;270:115130. https://doi.org/10.1016/J. APENERGY.2020.115130.
- [22] Merten M, Olk C, Schoeneberger I, Sauer DU. Bidding strategy for battery storage systems in the secondary control reserve market. Appl Energy Jun. 2020;268: 114951. https://doi.org/10.1016/J.APENERGY.2020.114951.
- [23] Olk C, Sauer DU, Merten M. Bidding strategy for a battery storage in the German secondary balancing power market. J Energy Storage Feb. 2019;21:787–800. https://doi.org/10.1016/J.EST.2019.01.019.
- [24] Rancilio G, Bovera F, Merlo M. Revenue stacking for BESS: fast frequency regulation and balancing market participation in Italy. Int Trans Electrical Energy Syst 2022;2022. https://doi.org/10.1155/2022/1894003.
- [25] Doenges K, Egido I, Sigrist L, Lobato Miguelez E, Rouco L. Improving AGC performance in power systems with regulation response accuracy margins using battery energy storage system (BESS). IEEE Trans Power Syst Jul. 2020;35(4): 2816–25. https://doi.org/10.1109/TPWRS.2019.2960450.
- [26] Lobato E, Sigrist L, Ortega A, González A, Fernández JM. Battery energy storage integration in wind farms: Economic viability in the Spanish market. Sustainable Energy Grids Networks Dec. 2022;32:100854. https://doi.org/10.1016/J. SEGAN.2022.100854.
- [27] Hameed Z, Træholt C, Hashemi S. Investigating the participation of battery energy storage systems in the Nordic ancillary services markets from a business perspective. J Energy Storage Feb. 2023;58:106464. https://doi.org/10.1016/J. EST.2022.106464.
- [28] AhmadiAhangar R, Plaum F, Haring T, Drovtar I, Korotko T, Rosin A. Impacts of grid-scale battery systems on power system operation, case of Baltic region. IET Smart Grid Apr. 2024;7(2):101–19. https://doi.org/10.1049/STG2.12142.
- [29] Lee D, Chiang Y, Chen YT, Tsai HH. Impacts of battery energy storage system on power grid smartness: case study of Taiwan power company. J Energy Storage May 2024;86:111188. https://doi.org/10.1016/J.EST.2024.111188.
- [30] Rancilio G, Vicario A, Merlo M, Berizzi A. Battery energy storage contribution to system frequency for grids with high renewable energy sources penetration. Power Syst Frequency Control: Modeling Adv Jan. 2023:133–50. https://doi.org/ 10.1016/B978-0-443-18426-0.00005-4.
- [31] Mousavi SMG, Nikdel M. Various battery models for various simulation studies and applications. Renew Sustain Energy Rev 2014;32:477–85. https://doi.org/ 10.1016/j.rser.2014.01.048.
- [32] Schmidt O, Melchior S, Hawkes A, Staffell I. Projecting the future Levelized cost of electricity storage technologies. Joule Jan. 2019;3(1):81–100. https://doi.org/ 10.1016/J.JOULE.2018.12.008.
- [33] Wu D, Ma X. Modeling and optimization methods for controlling and sizing Gridconnected energy storage: a review. Curr Sustainable/Renewable Energy Rep Jun. 2021;8(2):123–30. https://doi.org/10.1007/S40518-021-00181-9/TABLES/2.
- [34] Nebuloni R, et al. A hierarchical two-level MILP optimization model for the management of grid-connected BESS considering accurate physical model. Appl Energy 2023;334. https://doi.org/10.1016/j.apenergy.2023.120697.
- [35] Gatta FM, Geri A, Lauria S, Maccioni M, Palone F. Battery energy storage efficiency calculation including auxiliary losses: Technology comparison and operating strategies. in 2015 IEEE Eindhoven PowerTech Jun. 2015:1–6. https://doi.org/ 10.1109/PTC.2015.7232464.

- [36] Rancilio G, Merlo M, Lucas A, Kotsakis E, Delfanti M. BESS modeling: investigating the role of auxiliary system consumption in efficiency derating. In: in 2020 International Symposium on Power Electronics, Electrical Drives, Automation and Motion (SPEEDAM); Jun. 2020. p. 189–94. https://doi.org/10.1109/ SPEEDAM48782.2020.9161875.
- [37] Olis W, Rosewater D, Nguyen T, Byrne RH. Impact of heating and cooling loads on battery energy storage system sizing in extreme cold climates. Energy Sep. 2023; 278:127878. https://doi.org/10.1016/J.ENERGY.2023.127878.
- [38] Rancilio G, et al. Modeling a Large-Scale Battery Energy Storage System for Power Grid Application Analysis. Energies 2019;12(17). https://doi.org/10.3390/ en12173312.
- [39] European Commission. Commission regulation (EU) 2017/2195 establishing a guideline on the electricity balancing. Brussels; 2017.
- [40] Terna. SunSet Area Pubblica. Accessed: Feb. 24. [Online]. Available: https://m yterna.terna.it/SunSet/Public; 2022.
- [41] Rancilio G, Dimovski A, Bovera F, Moncecchi M, Falabretti D, Merlo M. Service stacking on residential BESS: RES integration by flexibility provision on ancillary services markets. Sustainable Energy Grids Networks Sep. 2023;35:101097. https://doi.org/10.1016/J.SEGAN.2023.101097.
- [42] Entso-e. Survey on ancillary services procurement balancing market design 2021. Brussels; 2022.
- [43] Ramboll. Ancillary services from new technologies. Vejle; 2019.
- [44] Entso-e. Single Intraday Coupling (SIDC). Accessed: Sep. 01. [Online]. Available: https://www.entsoe.eu/network_codes/cacm/implementation/sidc/; 2023.
- [45] P. Spodniak, K. Ollikka, S. Honkapuro, *The Relevance of Wholesale Electricity Market Places: The Nordic Case*. Valtion taloudellinen tutkimuskeskus, 2019. Accessed: Sep. 01, 2023. [Online]. Available: https://www.doria.fi/handle/10024/172539.
- [46] Terna. Codice di Rete, Allegato A15 : Partecipazione Alla Regolazione Di Frequenza e Frequenza-Potenza. In: Allegato A15; 2008.
- [47] WECC. Standard BAL-001-0.1a Real Power Balancing Control Performance. 2009.
 [48] ENTSO-E. TERRE. In: ENTSO-E; 2024. Accessed: Jul. 08. [Online]. Available:
- https://www.entsoe.eu/network_codes/eb/terre/. [49] Calise F, Cappiello FL, Dentice d'Accadia M, Vicidomini M. Smart grid energy
- [49] Cause F, Cappiello FL, Dentice of Accadia M, Vicidomini M. Smart grid energy district based on the integration of electric vehicles and combined heat and power generation. Energ Conver Manage Apr. 2021;234:113932. https://doi.org/ 10.1016/J.ENCONMAN.2021.113932.
- [50] Maier S. Smart energy systems for smart city districts: case study Reininghaus District. Energy Sustain Soc Dec. 2016;6(1):1–20. https://doi.org/10.1186/ S13705-016-0085-9/FIGURES/16.
- [51] Romero Rodríguez L, Salmerón Lissén JM, Sánchez Ramos J, Rodríguez Jara EÁ, Álvarez Domínguez S. Analysis of the economic feasibility and reduction of a building's energy consumption and emissions when integrating hybrid solar thermal/PV/micro-CHP systems. Appl Energy Mar. 2016;165:828–38. https://doi. org/10.1016/J.APENERGY.2015.12.080.
- [52] López Prol J, Steininger KW. Photovoltaic self-consumption regulation in Spain: profitability analysis and alternative regulation schemes. Energy Policy Sep. 2017; 108:742–54. https://doi.org/10.1016/J.ENPOL.2017.06.019.
- [53] M. M. Iqbal, I. Ali Sajjad, A. Manan, M. Waseem, A. Ali, A. Sohail, "Towards an Optimal Residential Home Energy Management in Presence of PV Generation, Energy Storage and Home to Grid Energy Exchange Framework," 2020 3rd international conference on computing, mathematics and engineering technologies: idea to innovation for building the knowledge economy, iCoMET 2020, Jan. 2020. doi: https://doi.org/10.1109/ICOMET48670.2020.9073798.
- 2020, Jan. 2020, doi: https://doi.org/10.1109/ICOMET48670.2020.9073798.
 [54] Dehler J, et al. Self-consumption of electricity from renewable sources. Europe's Energy Transition: Insights Policy Making Jan. 2017:225–36. https://doi.org/10.1016/B978-0-12-809806-6.00027-4.
- [55] Terna. Regolamento Fast Reserve. 2019.
- [56] Alizadeh MI, Parsa Moghaddam M, Amjady N, Siano P, Sheikh-El-Eslami MK. Flexibility in future power systems with high renewable penetration: a review. Renew Sustain Energy Rev May 2016;57:1186–93. https://doi.org/10.1016/J. RSER.2015.12.200.
- [57] Brusco G, Burgio A, Menniti D, Pinnarelli A, Sorrentino N. The economic viability of a feed-in tariff scheme that solely rewards self-consumption to promote the use of integrated photovoltaic battery systems. Appl Energy Dec. 2016;183:1075–85. https://doi.org/10.1016/J.APENERGY.2016.09.004.
- [58] Hassan Q, Abbas MK, Tabar VS, Tohidi S, Sameen AZ, Salman HM. Technoeconomic assessment of battery storage with photovoltaics for maximum selfconsumption. Energy Harvesting Syst Jan. 2024;11(1). https://doi.org/10.1515/ EHS-2022-0050/MACHINEREADABLECITATION/RIS.
- [59] di Milano Politecnico. Commissione Energia. Accessed: Feb. 28. [Online]. Available: https://www.commissionenergia.polimi.it/; 2022.
- [60] di Milano Politecnico. Piano Strategico di Sostenibilità 2023–2025. Milan; 2023.
 [61] European Commission. Photovoltaic Geographical Information System (PVGIS). Joint Research Centre; 2024. Accessed: Apr. 08. [Online]. Available: https://joint -research-centre.ec.europa.eu/photovoltaic-geographical-information-system-pvg is en.
- [62] Delfanti M, et al. Grid-Tie and Off-Grid Operations for an Innovative Microgrid Realized in Leonardo Campus of Politecnico di Milano. In: 5th International Forum on Research and Technologies for Society and Industry: Innovation to Shape the Future, RTSI 2019 - Proceedings; Sep. 2019. p. 103–8. https://doi.org/10.1109/ RTSI.2019.8895582.
- [63] TSOs. Explanatory document to the Amendment to the TSOs' Proposal for the establishment of common and harmonised rules and processes for the exchange and procurement of Balancing Capacity for Frequency Containment Reserves (FCR). 2021.

G. Rancilio et al.

- [64] Lee T. Energy storage in PJM: Exploring frequency regulation market transformation. Philadelphia; 2017.
- [65] Spiller M, et al. A Model-Aware Comprehensive Tool for Battery Energy Storage System Sizing. Energies 2023;16(18):6546. Sep. 2023, https://doi.org/10.3390/E N16186546.
- [66] Terna. Regolamento recante le modalità per la qualificazione e la fornitura del servizio di regolazione secondaria della frequenza/potenza tramite risorse non già abilitate. Rome; 2020.
- [67] ARERA. Approvazione del Testo Integrato del Dispacciamento Elettrico (TIDE). Milan; 2023. Accessed: Feb. 13, 2024. [Online]. Available: https://www.arera. it/atti-e-provvedimenti/dettaglio/23/345-23.
- [68] Figgener J, et al. The influence of frequency containment reserve flexibilization on the economics of electric vehicle fleet operation. J Energy Storage Sep. 2022;53: 105138. https://doi.org/10.1016/J.EST.2022.105138.